Author’s response to reviews

Title: Evaluation of Tp-e Interval, Tp-e/QT Ratio and Tp-e/QTc Ratio in Patients with Acute Myocarditis

Authors:
Fatih Mehmet Uçar (dr_fmucar@hotmail.com)
Cihan OZTURK (dr.cihanozturk@gmail.com)
Mustafa Adem YILMAZTEPE (mayilmaztepe@yahoo.com)

Version: 3 Date: 23 Aug 2019

Author’s response to reviews:

Dear Editor

Thank you for reading and reviewing our manuscript, which help us improve it to a better scientific level. We highly appreciated the constructive comments and valuable suggestions. We have modified the manuscript in response to your comments and addressed each of their concerns point-by-point as detailed below.

Editor Comments:
1. It has come to our attention that within your manuscript there is significant text overlap with other publications, particularly:

Introduction:

Materials and Methods:
Discussion:

Conclusion:

While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. Please re-phrase this section in your own words to avoid direct overlap (please note, we cannot accept direct textual overlap with any previously published works even if the authors of the work are yourselves). Please be informed that we cannot proceed with handling your manuscript before this issue is resolved, and the sections of text in question have been reformulated.

RESPONSE: Revised as your comments

2. Please format your abstract under the following headings:
   • Background
   • Methods
   • Results
   • Conclusions
   • Keywords

More information on what should be included in these sections can be found at https://bmcinfectedis.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

RESPONSE: Revised as your comments

3. Please change the section heading “Introduction” to “Background”.

RESPONSE: Revised as your comments

4. Please change the section heading “Materials and Methods” to “Methods”.

RESPONSE: Revised as your comments

5. In the Availability of data and materials section please clearly state who should be contacted if someone wants to request the data.

RESPONSE: Revised as your comments
6. The individual contributions of all authors to the manuscript should be specified in the Authors’ Contributions section. Additionally please clarify who the contributions of "FAE" and "SC" correspond to. Guidance and criteria for authorship can be found here:

http://www.biomedcentral.com/submissions/editorial-policies#authorship

RESPONSE: Revised as your comments

7. Please clarify whether any administrative permissions were required to access the raw data. If so please state who granted permission in the “Ethics approval and consent to participate” section.

RESPONSE: Revised as your comments

8. We note that you have not included an “Acknowledgements” section in the Declarations. Authors should obtain permission to acknowledge from all those mentioned in the “Acknowledgements” section. If no acknowledgements are necessary then please state 'None'.

RESPONSE: Acknowledgements section added to text

9. We note your Abstract has previously been presented elsewhere. Please ensure to state this in the Acknowledgements section of your Declarations.

https://academic.oup.com/europace/article/19/suppl_3/iii221/3872950

RESPONSE: Revised as your comments

10. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

RESPONSE: Revised as your comments